Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Oct. 7, 2009

Injector

· Injector was optimized: good and stable transmission, and three beams coincide. 
· Yesterday, Sept. 29, laser power meter was fixed. Laser spot was moved to boost the QE.

Accelerator

· Friday, Oct. 2, the north linac local oscillator 1427MHz amplifier failed, which tripped off all the RF (radio frequency) in the north linac; the faulty amplifier was replaced.
· Saturday, there were Capture cooling problems, and water leak in zone 2L21; the faulty parts were replaced. 
· Saturday the PSS dropped the injector to power due to a failed BCM -/+ 24VDC power supply; the power supply was replaced.
· Sunday, a vacuum valve in 4T region would not open due to a split air hose; the hose was repaired. 4T region orbits were not reproduced afterwards but Hall C beam has been stable.
· Run Krest to keep the energy spread low.

· The accelerator was running well.

· After yesterday commissioning of 1L04, three more cavities were back online. 

Hall A---E05-109; pass 3; energy 3.49 GeV
· Hall A beam is fine.
· Hall A is off line due to Hall C special energy runs.

· Hall A line steering was optimized to increase the aperture around ep region, which eliminated the ep ion chamber trip.

Hall B ---Eg6; pass 5; energy 5.764 GeV
· Hall B is back online now.
· Hall B current dropped to zero a few times. Possible causes: beam not stable, nA BPMs.

· Hall B may run up to 250nA, which exceed the Tagger dump and the FCup power limits. The hall has an insertable beam stopper right upstream the FCup. The stopper has 5kW capacity. For current higher than 30nA the stopper must be in to protect the FCup. A Chime has been set up to monitor the position of the stopper.

· The FSD node for BOM is now active to protect its target. A Chime will be set up to make sure its beam positions stay in range.

· Current lock monitors 2C24A not Fcup.

· Hall B is down for the experiment change over. Will be back online this Thursday swing shift. Everyone needs to read Hall B Experimental Binder.
Hall C---HKS; pass 2; energy 2,344 GeV
· Hall C beam is fine.
· Hall C energy spread read back is current dependant. The camera exposure time needs to be adjusted when current changes.

· The runs for the energy shifts were successful.

· Hall C special energy runs: 2344 + 1MeV, 2344 - 1MeV, and 2344 + 20 MeV starting today’s 99/30) owl shift.

· Hall C dump water pump failed Saturday. The secondary pump (with limited cooling capacity) was turned on. The faulty pump was repaired yesterday, Sept. 29.
· No issues with beam.

http://opweb.acc.jlab.org/CSUEApps/elog/elog.php
